Author's response to reviews

Title: Tamoxifen is not effective in good prognosis patients with hepatocellular carcinoma

Authors:

Ciro Gallo (ciro.gallo@unina2.it)
Ermelinda De Maio (linda.demaio@uosc.fondazionepascale.it)
Massimo Di Maio (massimo.dimaio@uosc.fondazionepascale.it)
Giuseppe Signoriello (Giuseppe.Signoriello@unina2.it)
Bruno Daniele (bruno.daniele@ao-rummo.it)
Sandro Pignata (sandro.pignata@fondazionepascale.it)
Annalisa Annunziata (Annlysa@yahoo.it)
Francesco Perrone (fr.perrone@agora.it)

Version: 2 Date: 30 June 2006

Author's response to reviews: see over
Naples, June 30th, 2006

Object: revised manuscript submission (MS: 1571994608102497)

Dear Editor,

We are submitting the revised version of the research article entitled “Tamoxifen is not effective in good prognosis patients with hepatocellular carcinoma” by Gallo et al.. We revised the manuscript in line with the comments provided by the third reviewer, better elucidating the question that gave rise to our report. We added a brief Conclusion after the Discussion and we clarified the contributions of all of the study's Authors. Furthermore, we have separated the figures (new version contains 4 figures instead of 2) and made them cropped.

Looking forward to hearing from you.

Sincerely yours,

Ciro Gallo MD

Department of Medicine and Public Health
Seconda Università di Napoli
Via L. Armanni 5
80138- Napoli, Italy
Phone: +39 0815666021 fax: +39 0815666021
e-mail: ciro.gallo@unina2.it